Author’s response to reviews

Title: Antiproliferative and apoptotic effects of proteins from black seeds (Nigella sativa) on human breast MCF-7 cancer cell line.

Authors:

Yamna Khurshid (yamnakhurshid@gmail.com)
Basir Syed (bsyed@chapman.edu)
Shabana Simjee (sh01us@hotmail.com)
Obaid Beg (obaidbeg@gmail.com)
Aftab Ahmed (aahmed@chapman.edu)

Version: 1 Date: 03 Sep 2019

Author’s response to reviews:

Nigella sativa seeds were bought from the local grocery store of Karachi, Pakistan and identified by Dr. Muneeba Khan, Taxonomist at Center for Plant Conservation, University of Karachi Herbarium and Botanical Garden.